Author's response to reviews

Title: A Qualitative Study of Factors Influencing Unavailability of Doctors and Nurses at Rural Healthcare Facilities in Bangladesh

Authors:

Emmanuel K Darkwa MPH (ekdarkwa@gmail.com)
M. Sophia Newman MPH (msophnewman@gmail.com)
Mahmuda Kawkab (kawkab@bracu.ac.bd)
Mahbub Elahi Chowdhury PhD (melahi@icddrb.org)

Version: 2 Date: 14 August 2014

Author's response to reviews: see over
Dear Editors,

Attached please find our response to an email regarding our study, “Factors Influencing Unavailability of Doctors and Nurses at Rural Healthcare Facilities in Bangladesh: A Qualitative Study.” We were instructed to include a competing interest statement and line and page numbering.

We have updated the document to include line and page numbering.

This Competing Interests statement was present in the correct spot in the manuscript on the first submission. We have altered it slightly to match your exact wording. It is highlighted.

We hope this addresses your concerns.

Yours,

Emmanuel K. Darkwa, MPH
On behalf of M. Sophia Newman, MPH, Mahmuda Kawkab, and Mahbub Elahi Chowdhury, PhD